Chohan /14--Nov-05
Week 46 starting  14 Nov 05
SM18 Operation Meeting (Tuesdays)

Bench status on Monday 14 Nov 05:  
 11/14/2005 3:55:51 PM
	9
	A1
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	295.18

	2
	A2
	MBAL3398
	Prep 5 Cool Down
	93.58

	4
	B1
	SSS277
	PT 13.1 IAP @ Warm
	300.82

	8
	B2
	SSS225
	ICS 2 Connect Magnet (ICS)
	294.1

	7
	C1
	MBAR3328
	Prep 3 CDW1 HV
	292.94

	5
	C2
	MBAL3414
	PT 12 Warm Up
	110.28

	6
	D1
	SSS123
	PT 12 Warm Up
	128.42

	12
	D2
	SSS193
	Prep 0 Scan Bench
	294.55

	10
	E1
	[image: image2.png]



	[image: image3.png]



	294.1

	3
	E2
	MBBL2140
	Thermal Cycle
	271.02

	11
	F1
	MBAL3139
	ICS 1 WP04 HV Test Warm
	292.94

	1
	F2
	MBBR1281
	Prep 5 Cool Down
	79.29


  Tempr. from TBSM18 site
 
   
Coordinator this week: John Miles; MTM support: 16-3602

 16 complete Magnet Tests cycles  11 Dipoles  + 4 SSS + 1 SSSS

 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 

Situation From Cryogenics:  J-P Lamboy or ACR Rep.

Special Points:
· Sunday 13/11  Rocla : Not possible to change anymore magnets !
· Lot of piling up & magnets all cooling down  at same time Monday onwards
· Lots of Tests on SSS277 ?

· Tests Programme for 3139 ?

Special SSS Q6 on D2 (J Halik  , Marco)
AOB 
	Magnets under Test from 06-11-2005 to 12-11-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR3390 
	[image: image4.png]



	


	05-11-2005 
	38 h.
	18 h.
	15 h.
	13 h.
	25 h.
	09-11-2005 
	109 h.
	Stripping
	
	

	2
	TBA2
	MBAL3412 
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	04-11-2005 
	22 h.
	19 h.
	22 h.
	16 h.
	19 h.
	08-11-2005 
	98 h.
	Stripping
	
	

	3
	TBA2
	MBAR2225 
	[image: image8.png]



	


	08-11-2005 
	26 h.
	21 h.
	24 h.
	17 h.
	11 h.
	12-11-2005 
	99 h.
	Stripping
	
	

	4
	TBB1
	SSS261 
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	04-11-2005 
	32 h.
	13 h.
	15 h.
	7 h.
	18 h.
	08-11-2005 
	85 h.
	Stripping
	
	

	5
	TBB2
	SSS033 
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	04-11-2005 
	33 h.
	16 h.
	16 h.
	9 h.
	17 h.
	09-11-2005 
	91 h.
	Stripping
	
	

	6
	TBC1
	MBBL3653 
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	04-11-2005 
	34 h.
	29 h.
	21 h.
	13 h.
	20 h.
	09-11-2005 
	117 h.
	Stripping
	
	

	7
	TBC2
	MBAR2231 
	[image: image16.png]



	


	31-10-2005 
	58 h.
	26 h.
	26 h.
	13 h.
	11 h.
	06-11-2005 
	134 h.
	Stand By
	
	

	8
	TBC2
	MBBR2241 
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	06-11-2005 
	54 h.
	19 h.
	28 h.
	12 h.
	7 h.
	11-11-2005 
	120 h.
	Stripping
	
	

	9
	TBD1
	SSS115 
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	07-11-2005 
	44 h.
	14 h.
	19 h.
	7 h.
	12 h.
	11-11-2005 
	96 h.
	Stripping
	
	

	10
	TBD1
	SSS218 
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	01-11-2005 
	35 h.
	31 h.
	27 h.
	11 h.
	38 h.
	07-11-2005 
	142 h.
	Stripping
	
	

	11
	TBD2
	SSS631 
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	25-10-2005 
	72 h.
	71 h.
	198 h.
	11 h.
	13 h.
	09-11-2005 
	365 h.
	Stand By
	
	

	12
	TBE1
	MBAR1273 
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	02-11-2005 
	26 h.
	30 h.
	16 h.
	15 h.
	31 h.
	07-11-2005 
	118 h.
	Stripping
	
	

	13
	TBE2
	MBAR1255 
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	01-11-2005 
	37 h.
	35 h.
	35 h.
	17 h.
	13 h.
	07-11-2005 
	137 h.
	Stripping
	
	

	14
	TBF1
	MBAR2248 
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	04-11-2005 
	30 h.
	22 h.
	29 h.
	14 h.
	14 h.
	09-11-2005 
	109 h.
	Stripping
	
	

	15
	TBF2
	MBAR2244 
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	01-11-2005 
	29 h.
	30 h.
	23 h.
	15 h.
	19 h.
	07-11-2005 
	116 h.
	Stripping
	
	

	16
	TBF2
	MBAR3376 
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	07-11-2005 
	37 h.
	23 h.
	42 h.
	16 h.
	7 h.
	12-11-2005 
	125 h.
	Stripping
	
	


	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR3390
	12614 A.
(D2_L)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3412
	11144.3 A.
(D2_U)
	11305 A.
(D1-U)
	12097.8 A.
(D2-U)
	12395 A.
(D1-L)
	12713.9 A.
(D2_U)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	MBAR2225
	11973.3 A.
(D2_L)
	12759.5 A.
(D1_U)
	12850 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS261
	12074 A.
(QE-34)
	12421.9 A.
(QI-34)
	12428.1 A.
(QI_34)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS033
	11755.7 A.
(QI_34)
	12390.3 A.
(QI_34)
	[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL3653
	10386.7 A.
(D1-U)
	11558.1 A.
(D2-U)
	12153.6 A.
(D1-L)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAR2231
	10811.4 A.
(D2_U)
	12032.3 A.
(D2_U)
	12232.9 A.
(D1-U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBR2241
	11721.5 A.
(D2_L)
	12741.3 A.
(D2-U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS115
	12139 A.
(QI_12)
	12137 A.
(QE-12)
	12464 A.
(QE_12)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS218
	12010 A.
(QI_12)
	12477.3 A.
(QE-12)
	12639 A.
(QI-34)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	SSS631
	5422 A.
(E_P12)
	5815 A.
(E_P34)
	5820 A.
[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR1273
	12505 A.
(D1-L)
	12850 A.
[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAR1255
	11547.6 A.
(D1-U)
	12182.6 A.
(D2_U)
	12457.3 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAR2248
	11954.5 A.
(D1_L)
	12792.3 A.
(D1-L)
	12656 A.
(D1-L)
	12850 A.
[image: image49.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBAR2244
	10967.9 A.
(D1_U)
	12520.7 A.
(D2_U)
	12850 A.
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	12850 A.
[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBAR3376
	12643 A.
(D2_L_S01)
	12863 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_08_A
	TBF1


	MBAR 2248: TT821 reading during Powering tests is 1.552 K. So NC generated in CDW2.
	MTM
	
	MB was already warm. To react efficiently equip.support has to be called when the magnet is cold.

	Wednesday_09_A
	TBE1


	Polarity switch difficult to handle in left position.
	MTM
	
	Giorgio called and solved the problem.

	Wednesday_09_N
	TBC2


	Quench during the de-ramp of the current after SPA test at 12000A.
	AB/PO
	
	Report to AB/PO

	Thursday_10_M
	TBF2


	MM1 on MB3376:: MMP software does not recognise the magnet name. Unable to proceed with MM1. 
	AB/CO
	
	Federico was informed. Problem solved. MMPvi was corrupted.

	Thursday_10_A
	TBF2
	Lost of cryo OK for all benches
	ACR
	
	

	Thursday_10_A
	TBF2
	LHC cycle on MB3376 stopped because of a SPA.
	ACR
	
	Origin not known

	Thursday_10_A
	TBF2


	Ramp test launched. Stopped because current reading through MMP is not working in Ap1
	AB/CO
	
	Federico called and changed the DIGIO card.

	Thursday_10_N
	TBF2


	Cryo lost during LHC cycle.
	ACR
	
	

	Friday_11_M
	TBA2
	MB2225: SPA occurred during the plateau at 12850 A before generating the u_file.
	AB/PO
	
	Origin not knownTest will be repeated. 

	Friday_11_A
	TBE1
	Quench during PT8 performed on MB2179
	AB/PO
	
	TC noticed that this test was performed in the first run so he advised to proceed with the tests. 

	Friday_11_N
	TBF2
	ROCLA parked in front of the bench impeded ICS team too remove the shafts from MB3376
	ICS
	
	ICS team tried to contact ROCLA team.

	Friday_11_N
	TBB1


	Quench on SSS277: TC asked to perform 2 quenches at 2.3 K
	AB/CO
	
	ACR team answered that it is not possible at this temperature.

	Saturday_12_A
	TBE2


	During de-ramp of the current on MB2140 after PT8 quench at 1340 A.
	AB/PO
	
	Tripping of the power supply? TC decided to repeat the test.

	Saturday_12_A
	TBD2


	The Heat Sink of the Orbit Dipole connector no. 4 is bent - BEFORE connection by OP!
	
	
	

	Sunday_13_N
	TBD1


	SSS123: Signal for the external resistor QH3 is displayed flat from hfoff (AQA QH viewer). Signal are OK from lf application
	AB/CO?
	Hf (AQA)
	Federico advised to proceed with the tests.
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	Wednesday_09_A
	TBE1


	Polarity switch difficult to handle in left position.
	MTM
	
	Giorgio called and solved the problem.

	Wednesday_09_N
	TBC2


	Quench during the de-ramp of the current after SPA test at 12000A.
	AB/PO
	
	Report to AB/PO

	Thursday_10_M
	TBF2


	MM1 on MB3376:: MMP software does not recognise the magnet name. Unable to proceed with MM1. 
	AB/CO
	
	Federico was informed. Problem solved. MMPvi was corrupted.

	Thursday_10_A
	TBF2
	Lost of cryo OK for all benches
	ACR
	
	

	Thursday_10_A
	TBF2
	LHC cycle on MB3376 stopped because of a SPA.
	ACR
	
	Origin not known

	Thursday_10_A
	TBF2


	Ramp test launched. Stopped because current reading through MMP is not working in Ap1
	AB/CO
	
	Federico called and changed the DIGIO card.

	Thursday_10_N
	TBF2


	Cryo lost during LHC cycle.
	ACR
	
	

	Friday_11_M
	TBA2
	MB2225: SPA occurred during the plateau at 12850 A before generating the u_file.
	AB/PO
	
	Origin not knownTest will be repeated. 

	Friday_11_A
	TBE1
	Quench during PT8 performed on MB2179
	AB/PO
	
	TC noticed that this test was performed in the first run so he advised to proceed with the tests. 

	Friday_11_N
	TBF2
	ROCLA parked in front of the bench impeded ICS team too remove the shafts from MB3376
	ICS
	
	ICS team tried to contact ROCLA team.

	Friday_11_N
	TBB1


	Quench on SSS277: TC asked to perform 2 quenches at 2.3 K
	AB/CO
	
	ACR team answered that it is not possible at this temperature.

	Saturday_12_A
	TBE2


	During de-ramp of the current on MB2140 after PT8 quench at 1340 A.
	AB/PO
	
	Tripping of the power supply? TC decided to repeat the test.

	Saturday_12_A
	TBD2


	The Heat Sink of the Orbit Dipole connector no. 4 is bent - BEFORE connection by OP!
	
	
	

	Sunday_13_N
	TBD1


	SSS123: Signal for the external resistor QH3 is displayed flat from hfoff (AQA QH viewer). Signal are OK from lf application
	AB/CO?
	Hf (AQA)
	Federico advised to proceed with the tests.


